Author’s response to reviews

Title: Trends in influenza vaccination uptake among people aged over 74 years, 1997-2000: Survey of 73 general practices in Britain

Authors:

Ms Elizabeth Breeze (elizabeth.breeze@lshtm.ac.uk)
Dr Punam Mangtani (punam.mangtani@lshtm.ac.uk)
Gill M Price (gill.price@lshtm.ac.uk)
Sari Kovats (sari.kovats@lshtm.ac.uk)
Jenny Roberts (jenny.roberts@lshtm.ac.uk)
Prof Astrid E Fletcher (astrid.fletcher@lshtm.ac.uk)

Version: 3 Date: 7 Apr 2004

Dear Editor

1762290220240945 second revision 5 April 2004

Thank you for your email of 26 March. All the changes suggested there have been made. The Mangtani reference is in Press still.

In addition the following changes were made:

Page 3 Background para1. Last sentence adapted to incorporate both the new and old references (now references 5 and 6)
Page 5. Analysis strategy para 1. Reference to Stata software added.

Tables 2 and 3 have been re-formatted to make them fit the page better.

Please let us know if there are any further changes needed.

Yours sincerely,

Elizabeth Breeze